Author's response to reviews

Title: Study protocol: Insulin and its role in cancer

Authors:

K Harish (drkhari@yahoo.com)
M Dharmalingam (mala_endo@rediffmail.com)
M Himanshu (himanshu.yajaman@gmail.com)

Version: 2 Date: 20 October 2007

Author's response to reviews: see over
From,
Dr Harish K
# 2866, 13th Main, ‘E’ Block
Subramanyanagar
Bangalore – 560010
INDIA

To,
The Editor/s
BMC Endocrine Disorders

Sir,

The questions raised on competing financial interests were:

“In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.

Do you have any other financial competing interests? If so, please specify.

Non-financial competing interests
Are there any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.”

The answer for all these questions are “NO”

All the desired changes have been complied with point by point.

We are resubmitting the manuscript with the desired changes.

Awaiting your favorable reply

Thanking you

Yours sincerely

Harish K